Author's response to reviews

Title: Eye care utilization patterns in Tehran population: a population based cross-sectional study

Authors:

Akbar Fotouhi (afotouhi@sina.tums.ac.ir)
Hassan Hashemi (hhashemi@tums.ac.ir)
Kazem Mohammad (mohamadk@tums.ac.ir)

Version: 5 Date: 7 January 2006

Author's response to reviews:

In the manuscript text, I placed the reference before the full stop.